

# **Geophysical Research Letters**<sup>®</sup>

## **RESEARCH LETTER**

10.1029/2024GL112446

#### **Key Points:**

- Subregions of the West Antarctic Peninsula marine system show varying amplitudes of seasonal cycles of air-sea CO<sub>2</sub> flux and Chlorophyll-a
- Seasonal cycles of both air-sea CO<sub>2</sub> flux and Chlorophyll-a increase in amplitude moving poleward
- Massive biological drawdown of CO<sub>2</sub> occurs over the Southern Shelf in summer

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

J. S. Turner, jturners@odu.edu

#### Citation:

Turner, J. S., Munro, D. R., Fay, A., Stammerjohn, S., Kim, H., Schofield, O., & Dierssen, H. (2025). Seasonal variability of surface ocean carbon uptake and Chlorophyll-a concentration in the West Antarctic Peninsula over two decades. *Geophysical Research Letters*, *52*, e2024GL112446. https://doi.org/10.1029/ 2024GL112446

Received 7 SEP 2024 Accepted 23 DEC 2024

© 2025. The Author(s). This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Seasonal Variability of Surface Ocean Carbon Uptake and Chlorophyll-a Concentration in the West Antarctic Peninsula Over Two Decades

Jessica S. Turner<sup>1,2</sup>, David R. Munro<sup>3,4</sup>, Amanda Fay<sup>5</sup>, Sharon Stammerjohn<sup>6</sup>, Heather Kim<sup>7</sup>, Oscar Schofield<sup>8</sup>, and Heidi Dierssen<sup>1</sup>

<sup>1</sup>Department of Marine Sciences, University of Connecticut, Groton, CT, USA, <sup>2</sup>Department of Ocean and Earth Sciences, Old Dominion University, Norfolk, VA, USA, <sup>3</sup>Global Monitoring Laboratory, National Oceanic and Atmospheric Administration, Boulder, CO, USA, <sup>4</sup>Cooperative Institute for Research in Environmental Sciences, University of Colorado, Boulder, CO, USA, <sup>5</sup>Columbia University and Lamont-Doherty Earth Observatory, Palisades, NY, USA, <sup>6</sup>Institute of Arctic and Alpine Research, University of Colorado, Boulder, CO, USA, <sup>7</sup>Department of Marine Chemistry and Geochemistry, Woods Hole Oceanographic Institution, Woods Hole, MA, USA, <sup>8</sup>Department of Marine and Coastal Sciences, Rutgers University, New Brunswick, NJ, USA

**Abstract** The Southern Ocean plays a vital role in global  $CO_2$  uptake, but the magnitude and even the sign of the flux remain uncertain, and the influence of phytoplankton phenology is underexplored. This study focuses on the West Antarctic Peninsula, a region experiencing rapid climate change, to examine shifts in seasonal carbon uptake. Using 20 years of in situ air-sea  $CO_2$  flux and satellite-derived Chlorophyll-a, we observe that the seasonal cycles of both air-sea  $CO_2$  flux and Chlorophyll-a intensify poleward. The amplitude of the seasonal cycle of the non-thermal component of surface ocean p $CO_2$  increases with increasing latitude, while the amplitude of the thermal component remains relatively stable. Pronounced biological uptake occurs over the shelf in austral summer despite reduced  $CO_2$  solubility in warmer waters, which typically limits carbon uptake through physical processes. These findings underscore the prominence of biological mechanisms in regulating carbon fluxes in this rapidly changing region.

**Plain Language Summary** The Southern Ocean plays a key role in absorbing carbon dioxide from the atmosphere, but we are still trying to understand exactly how much carbon dioxide the Southern Ocean absorbs and what factors influence this process. The West Antarctic Peninsula is an important area for studying these changes because it is experiencing rapid warming. We looked at changes in the amount of carbon dioxide in the surface ocean water and the amount of microscopic drifting plants (phytoplankton) based on the amount of the pigment Chlorophyll-a over the past 20 years. Seasonal cycles of both carbon dioxide and Chlorophyll-a increase from north to south in their amplitude, or their range between the minimum and the maximum each year. In summer, contrary to the effect of warm temperatures which would usually limit carbon uptake by way of physics, strong biological drawdown by phytoplankton dominates the signal.

## 1. Introduction

Despite the Southern Ocean playing a major role in the global carbon budget, many questions remain about the role of phytoplankton as a mechanism for carbon uptake. The Southern Ocean acts as a central hub for the world's oceans, where waters from all the world's water masses converge and mix. It is a region of deep water formation and represents an important region in regulating pre-industrial and anthropogenic ocean carbon levels, thus also impacting atmospheric  $CO_2$  levels (Caldeira & Duffy, 2000; Marinov et al., 2006; Sarmiento & Orr, 1991). Debate remains about whether the Southern Ocean is a net sink or source for atmospheric  $CO_2$ . Models and airborne observations suggest that the Southern Ocean is a  $CO_2$  net sink (Long et al., 2021), but early results from profiling floats suggest that it is a net source (Bushinsky et al., 2019; Gray et al., 2018). The magnitude of the difference between those two estimates is ~0.9 Gt C yr<sup>-1</sup> (Long et al., 2021), approximately one third of the total global air-sea  $CO_2$  flux ( $\Delta pCO_2$ ) (2.8 Gt C yr<sup>-1</sup>) (Friedlingstein et al., 2023). Limited observations due to its remote location and rough conditions are the main reason for this uncertainty.

The role of phytoplankton processes is often underemphasized despite its contribution to  $\Delta pCO_2$ , specifically in the Southern Ocean. Instead, physical mechanisms are more commonly emphasized, such as isopycnal transport,

diapycnal eddy diffusion, transfer from the continental shelf to the deep ocean, entrainment/detrainment, and airsea gas exchange (Arroyo et al., 2020; Bernardello et al., 2014; Caldeira & Duffy, 2000; Yang et al., 2021). However, in many regions of the Southern Ocean, strong summer season primary production is a key contributor to the seasonal oceanic  $CO_2$  sink (Carrillo et al., 2004; Legge et al., 2015, 2017; Schultz et al., 2021; Tortell et al., 2015). Compared with other oceans, the Southern Ocean is unique in that biology plays a dominant role in the surface ocean carbon cycle on all timescales (Fay & McKinley, 2017). The biological contribution of phytoplankton production, while partially balanced by physical mechanisms at annual timescales (Munro, Lovenduski, Stephens, et al., 2015), is a principal factor, especially within the seasonal sea ice zone.

Regions vary widely and the Southern Ocean cannot be considered as a single ecosystem. Phytoplankton from different zones respond differently to environmental forcings over time (Turner et al., 2024) with polynyas and coastal zones often acting as substantial carbon sinks despite their small size (Monteiro, Kerr, & Machado, 2020; St-Laurent et al., 2019). The Drake Passage and continental shelf west of the Antarctic Peninsula (Figure 1) serve as a representative case study of spatial variability in phytoplankton biomass, phenology, and air-sea carbon exchange. The Drake Passage is a highly variable region in terms of carbon uptake, with multiple zones of  $CO_2$  sinks and sources delimited by marine fronts (Arbilla et al., 2024). Drake Passage Chlorophyll-a concentration (Chl-a) is relatively low compared to coastal Antarctic waters, representative of conditions in the broader subpolar Southern Ocean (Fay et al., 2018).

In contrast, over the continental shelf, the seasonal cycle of air-sea carbon exchange is larger in amplitude, with surface ocean Chl-a and carbon uptake generally increasing moving poleward due to biological processes (Eveleth et al., 2017; Kim et al., 2018; Turner et al., 2024). Within the West Antarctic Peninsula (WAP) region, ocean biogeochemistry and physics are highly regionalized (Testa et al., 2021), and there is high spatial variability in carbonate system parameters in the Mid Shelf subregion (Hauri et al., 2015). Mooring data from the WAP continental shelf show that this area is an annual atmospheric  $CO_2$  sink, with substantial biological drawdown in spring and summer only partially replenished by physical processes since seasonal sea ice suppresses winter outgassing (Shadwick et al., 2021; Yang et al., 2021). The WAP region is a sentinel region for polar ecosystem change with regionally relevant sinks for anthropogenic  $CO_2$  (Arrigo et al., 2008; Henley et al., 2019; Schofield et al., 2018). The WAP region is losing sea ice more rapidly than other parts of Antarctica (Reid et al., 2024; Stammerjohn & Scambos, 2020), and that loss impacts water column stability, primary production and surface ocean carbon uptake (Brown et al., 2019; Ducklow et al., 2013; Schofield et al., 2017).

When considering the contribution of primary production to the seasonal  $CO_2$  sink, the seasonal timing or phenology is often ignored. Shifts in the timing of the occurrence of the phytoplankton growing season have the potential to alter carbon cycling. Most observations in Antarctica are made in summer, thus seasonal dynamics are understudied and longer-term fluctuations in the rest of the seasonal cycle are mostly unknown. Phenological analysis is especially crucial in this region (Cimino et al., 2023) because most of the WAP shows high interannual variability in the seasonal cycle of Chl-a via satellite observations, with interannual variability increasing over time (Thomalla et al., 2011, 2023). Ship-based observations of the entire WAP marine ecosystem are available during summer and there are year-round mooring observations of pCO<sub>2</sub> for some isolated years; however, satellite Chl-a and ship-based pCO<sub>2</sub> observations specifically enable analysis of the seasonal cycle of the surface ocean carbon cycle and its biological component. The consequences of seasonality need to be considered.

This paper documents strong north-to-south gradients in the amplitude of seasonal cycles of  $\Delta pCO_2$  and Chl-a and demonstrates tight coupling between  $\Delta pCO_2$  and Chl-a seasonal cycles indicating the strong seasonal biological drawdown in this region. These findings contribute to resolving the ongoing debate about whether the Southern Ocean is a net sink or source of atmospheric CO<sub>2</sub>, a question with significant implications for global carbon budget estimates.

### 2. Methods

2.1. Data Sources

### 2.1.1. In Situ $\Delta pCO_2$ , $pCO_{2atm}$ , and $pCO_{2sur}$ Observations

 $\Delta pCO_2$  was calculated using the observed partial pressure of CO<sub>2</sub> in the atmosphere (pCO<sub>2atm</sub>) and partial pressure of CO<sub>2</sub> in the surface ocean (pCO<sub>2sur</sub>) as the difference pCO<sub>2sur</sub>-pCO<sub>2atm</sub>. pCO<sub>2atm</sub> is the dry air mixing ratio of atmospheric CO<sub>2</sub> from the Global Monitoring Laboratory surface marine boundary layer CO<sub>2</sub> (Lan



# **Geophysical Research Letters**



Figure 1. Map of the study area  $(55-80^{\circ}W, 57-70^{\circ}S)$  with subregions Drake Passage (yellow), Northern Shelf (red), Mid Shelf (green), and Southern Shelf (cyan), showing (a) bathymetry, (b) mean satellite-derived Chl-a 1997–2022, and (c) ship track locations for underway pCO<sub>2</sub> data 2000–2020.

et al., 2023) multiplied by NCEP sea level pressure (Kalnay et al., 1996) at monthly resolution and applying the water vapor correction according to Dickson et al. (2007). Underway  $pCO_{2sur}$  data were collected on research vessels (ship tracks) from 2000 to 2020 (Figure 1a).  $pCO_{2sur}$  was collected mainly from ships using underway and discrete sampling. More detailed methods for the collection of data in the Drake Passage Time-series and Surface Ocean CO<sub>2</sub> Atlas (SOCAT) v2023 (Bakker et al., 2016) are described in detail by Munro, Lovenduski, Stephens et al. (2015), Munro, Lovenduski, Takahashi et al. (2015). Many of the in situ  $pCO_{2sur}$  observations were

collected as part of the Palmer Long Term Ecological Research (Pal-LTER) program.  $pCO_{2sur}$  observations are collected in nearly every month of the year, which is uncommon in the Southern Ocean region, and the high sampling frequency makes this seasonal analysis possible.

#### 2.1.2. Satellite Chl-a Observations

Satellite-derived Chl-a data (Turner, 2025) were sourced from OC-CCI v6.0 (Sathyendranath et al., 2019), a merged multi-sensor, 4 km spatial resolution, monthly level-3 product using a global Chl-a algorithm estimated using a blended combination of OCI, OCI2, OC2, and OCX algorithms depending on optical water class membership (Gohin et al., 2002; Hu et al., 2012; O'Reilly & Werdell, 2019). Satellite-derived surface Chl-a is useful in polar regions despite inherent limitations (e.g., imperfect metric for biomass, high solar zenith angle, surface-only, limited winter data) because satellite data availability enables analysis over austral spring, summer, and early fall in the WAP (e.g., Turner et al., 2024). Ocean color data were used where >15% of monthly observations were present to avoid analyzing locations covered by sea ice for most of the record. Since global algorithms underestimate in situ WAP Chl-a by a factor of 2–2.5 (Dierssen & Smith, 2000; Kahru & Mitchell, 2010; Mitchell, 1992; Mitchell & Holm-Hansen, 1991), a correction was applied to better reflect the spatial range of in situ Chl-a for the WAP region as in Dierssen and Smith (2000), which had minimal effects at low Chl-a concentrations representative of offshore waters where the global algorithm generally performs well (Dierssen, 2000; Haëntjens et al., 2017) (Figure S1 in Supporting Information S1).

#### 2.2. Data Analysis

Spatially, four subregions of the WAP were analyzed (Figure 1). The Drake Passage region encompasses the narrow ocean region which includes the Drake Passage Time-series ship tracks. The shelf regions are bounded by the edge of the shelf (1,000 m isobath) and sub-divided into Northern, Mid, and Southern Shelf subregions based on long-term mean Chl-a concentrations. The Drake Passage subregion here corresponds to zones from Gray et al. (2018), Haëntjens et al. (2017), and Fay and McKinley (2014) which have similar characteristics in terms of temperature, Chl-a, and seasonality. The Northern Shelf subregion corresponds to an area north of the traditional area sampled every austral summer by the Pal-LTER program, whereas the Mid Shelf subregion used here corresponds to main area sampled by the Pal-LTER program. The Southern Shelf subregion includes Marguerite Bay and comprises the southernmost portion of the Pal-LTER sampling area.

Temporally, monthly means were used to mitigate data sparsity especially of in situ  $pCO_{2sur}$  observations (Figure S2, Table S1 in Supporting Information S1). Satellite Chl-a data are unavailable over the shelf from April to September due to low light and the presence of sea ice. Ship-based  $pCO_{2sur}$  observations are limited for the winter season in the Mid Shelf and slightly biased toward summer for all subregions. Data were analyzed for austral years 2000–2020 that is, 1 July 2000 to 30 June 2020. Seasons used for the Southern Hemisphere are delimited by the months December to February (summer), March to May (fall), June to August (winter), and September to November (spring).

Sea surface temperature (SST) values from concurrent underway ship track data were used to calculate thermally driven ( $pCO_{2sur}$ -T) and non-thermally-driven ( $pCO_{2sur}$ -non-T) components of the  $pCO_{2sur}$  seasonal cycles.  $pCO_{2sur}$ -T and  $pCO_{2sur}$ -non-T components were calculated following the approach of Takahashi et al. (1993) using equations from Wanninkhof et al. (2022).

Correlations between seasonal cycles of Chl-a and pCO<sub>2</sub> seasonal cycles ( $\Delta$ pCO<sub>2</sub>, pCO<sub>2sur</sub>, pCO<sub>2sur</sub>-T, and pCO<sub>2sur</sub>-non-T) were calculated over all months per year when data were present. We determined the Pearson correlation coefficient (*R*) between variables and computed the significance (*p*) by testing the null hypothesis that the correlation is zero using a Student's t-distribution via MATLAB function "corrcoef". Amplitudes of seasonal cycles were calculated via the difference between the maximum and minimum monthly values over the year for each decadal mean seasonal cycle (i.e., for 2000–2010, Chl-a seasonal cycle amplitude = maximum in January-minimum in September).



# **Geophysical Research Letters**



**Figure 2.** Seasonal cycles of Chl-a,  $\Delta pCO_2$ , and  $pCO_{2sur}$  with thermal and non-thermal components 2000–2020 showing Chl-a (a–d),  $\Delta pCO_2$  (e–h), and  $pCO_{2sur}$  (i–l) for the Drake Passage (a, e and i), Northern Shelf (b, f and j), Mid Shelf (c, g and k), and Southern Shelf (d,h,l) subregions. (See Figure S3 in Supporting Information S1 for  $\Delta pCO_2$  as in e-h with regionally scaled *y*-axes). Error bars represent ±1 standard deviation.

### 3. Results

### 3.1. Seasonal Patterns and Amplitudes of Seasonal Cycles

Seasonal cycles in Chl-a and  $\Delta pCO_2$  increase in amplitude moving poleward (Figure 2, Table 1). In the Drake Passage, Chl-a is generally low (<1 mg m<sup>-3</sup>) with a small amplitude seasonal cycle (0.5 mg m<sup>-3</sup>) (Figure 2a). The  $\Delta pCO_2$  seasonal cycle in the Drake Passage is similarly relatively small in amplitude (18 µatm) (Figure 2e, Figure S3 in Supporting Information S1). In the Northern Shelf, the Chl-a seasonal cycle amplitude of 1.2 mg m<sup>-3</sup> (Figure 2b) and the  $\Delta pCO_2$  seasonal cycle amplitude of 89 µatm (Figure 2f) are larger than those of the Drake Passage but not as substantial as those in subregions farther to the south. In the Mid Shelf, Chl-a shows a greater seasonal cycle amplitude of 1.8 mg m<sup>-3</sup> (Figure 2c) and a similarly large  $\Delta pCO_2$  seasonal cycle amplitude (143 µatm) (Figure 2g). The Southern Shelf has the highest amplitude seasonal cycles of any subregion for both Chl-a (4.5 mg m<sup>-3</sup>) (Figure 2d) and  $\Delta pCO_2$  (186 µatm) (Figure 2h).

The thermal and non-thermal components of the pCO<sub>2sur</sub> seasonal cycle are balanced in the Drake Passage, but non-thermal drivers including biology overpower the seasonal cycle over the shelf, particularly in the Southern Shelf subregion. The small amplitudes of the pCO<sub>2sur</sub> and  $\Delta$ pCO<sub>2</sub> seasonal cycles in the Drake Passage result from the opposing effects of temperature and non-temperature related factors including biology (Figure 2i). Winter cooling decreases pCO<sub>2sur</sub> concurrent with deep mixing increasing pCO<sub>2sur</sub>. Summer warming increases pCO<sub>2sur</sub> while biological carbon uptake lowers pCO<sub>2sur</sub>. In the Drake Passage, these components balance one another to result in a small-amplitude seasonal cycle. In the Northern Shelf and Mid Shelf, the thermal component is only partially balanced by the non-thermal component, with the non-thermal component showing strong



## Table 1

Amplitudes and Correlations of Chl-a,  $\Delta pCO_2$ ,  $pCO_{2sur}$  and the Thermal and Non-Thermal Components of  $pCO_{2sur}$  Seasonal Cycles

	Drake passage		Northern shelf		Mid shelf		Southern shelf	
Seasonal Cycle Amplitudes								
Chl-a	$0.5 \text{ mg m}^{-3}$		$1.2 \text{ mg m}^{-3}$		$1.8 \text{ mg m}^{-3}$		$4.5 \text{ mg m}^{-3}$	
$\Delta pCO_2$	18 µatm		89 µatm		143 µatm		186 µatm	
pCO <sub>2sur</sub>	20 µatm		93 µatm		145 µatm		195 µatm	
pCO <sub>2sur</sub> -T	66 µatm		51 µatm		47 µatm		35 µatm	
pCO <sub>2sur</sub> -non-T	76 µatm		139 µatm		182 µatm		224 µatm	
Seasonal Cycle Correlations <sup>a</sup>								
	R <sup>a</sup>	р	R	р	R	р	R	р
Chl-a versus $\Delta pCO_2$	-0.76	0.018	-0.97	< 0.001	-0.93	0.001	-0.95	0.001
Chl-a versus pCO <sub>2sur</sub>	-0.66	0.055	-0.97	< 0.001	-0.93	0.001	-0.95	0.001
Chl-a versus pCO <sub>2sur</sub> -T	-0.07	0.854	0.96	< 0.001	0.90	0.002	0.90	0.006
Chl-a versus pCO <sub>2sur</sub> -non-T	-0.13	0.740	-0.98	< 0.001	-0.94	0.000	-0.95	0.001

*Note.* Italicized values indicate significant correlations (p < 0.05). <sup>a</sup>Pearson correlation coefficient.

negative correlations with the Chl-a seasonal cycle (Table 1). In the Southern Shelf, the non-thermal component dominates the seasonal cycle (Figure 21). This far south, biological production is strong enough to substantially overbalance the thermal component.

#### 3.2. Coupling Between Seasonal Cycles of Chl-a and $\Delta pCO_2$

Seasonal cycles of phytoplankton biomass and carbon uptake are generally the inverse of one another in subpolar and polar regions (Körtzinger et al., 2008; Lüger et al., 2004; Takahashi et al., 1993, 2002). In this analysis, Chl-a and  $\Delta pCO_2$  seasonal cycles are strongly and significantly negatively correlated to one another (p < 0.05) in every subregion (Figure 2, Table 1). The strongest Chl-a– $\Delta pCO_2$  correlations occur in the three shelf subregions (R = -0.97, -0.93, and -0.95). Slightly less strong Chl-a– $\Delta pCO_2$  correlations are found in the Drake Passage (R = -0.76) where the seasonal cycle of  $\Delta pCO_2$  is less distinct. Chl-a and  $pCO_{2sur}$  seasonal cycles are strongly negatively correlated over the shelf, but not significantly in the Drake Passage. Over the shelf, correlations between Chl-a and the thermal and non-thermal components of  $pCO_{2sur}$  are strongly correlated, with positive correlations (R = 0.9 to 0.96) between the Chl-a and the thermal component and negative correlations (R = -0.94to -0.98) between Chl-a and the non-thermal component (which includes biology). In the Drake Passage, no correlation was found between Chl-a and the thermal and non-thermal components of  $pCO_{2sur}$  (Table 1).

#### 4. Discussion

#### 4.1. Biologically Mediated $\Delta pCO_2$ Seasonal Cycles

The seasonal cycle of  $pCO_{2sur}$  in the subregions is driven by a combination of thermal and non-thermal factors, with non-thermal components (including biological processes) showing larger amplitudes than thermal components and stronger correlations with Chl-a (Table 1, Figure 2). Sea ice formation, meltwater, and vertical mixing likely regulate pCO<sub>2</sub> seasonal cycles in addition to temperature and biological influences; however, the strong correlations between Chl-a and the non-thermal component of the pCO<sub>2</sub> seasonal cycles support the idea that biological production is a strong driver over the shelf. In the Drake Passage, the opposing influences of temperature and biology result in a small-amplitude seasonal cycle and a modest CO<sub>2</sub> sink. Conversely, in the Northern and Mid Shelves, the non-thermal factors, including biological production, producing a large seasonal cycle amplitude and functioning as a noteworthy CO<sub>2</sub> sink.

These results suggest that biology dominates the seasonal cycle of surface ocean carbon uptake, especially over the Southern Shelf. Correlations between the seasonal cycles of  $\Delta pCO_2$  and Chl-a are strongly negative in all

subregions (Figure 2, Table 1). The amplitudes of the non-thermally-driven component of the seasonal cycle of pCO<sub>2sur</sub> (which includes biology) are larger in all subregions than those of the thermally driven component (Figures 2i–2l). pCO<sub>2sur</sub>-non-T amplitudes reach 224 µatm (Southern Shelf), while pCO<sub>2sur</sub>-T seasonal cycle amplitudes range from 35 to 66 µatm (Table 1). The thermal component of the seasonal cycle is mirrored in SST seasonal cycles, which show warm temperatures in summer months with the highest temperatures occurring in summer in all subregions. Contrary to the effect of summer warming which would typically limit carbon uptake via low solubility in other ocean regions, strong biological drawdown dominates the  $\Delta pCO_2$  seasonal signal during these warm months in the shelf subregions. The Southern Ocean is the only global ocean region where biology controls the surface ocean carbon cycle at all timescales (Fay & McKinley, 2017); the WAP continental shelf is a prime example of this phenomenon. Biology-driven control of pCO<sub>2</sub> has been observed in the WAP more so than in other subregions of the Pacific Southern Ocean (Mo et al., 2023), and in the Northern Shelf biological activity was a dominant surface ocean carbon removal mechanism (Ito et al., 2018). The Drake Passage, with its open ocean characteristics, allows physical mechanisms to partially balance out biological uptake, acting to dampen the seasonal cycle in this region (Munro, Lovenduski, Stephens et al., 2015). Conversely, over the WAP continental shelf, the seasonal cycles of  $\Delta pCO_2$  and Chl-a are tightly coupled (Figure 2, Table 1) and large in amplitude, suggesting that biology governs the surface ocean carbon cycle over the shelf.

#### 4.2. Strong North–South Gradient in Biological Carbon Drawdown

The amplitudes of the seasonal cycles of  $\Delta pCO_2$  and Chl-a increase moving poleward (Figure 2). The Drake Passage  $\Delta pCO_2$  seasonal cycle amplitude of 18 µatm compares well with that of the Southern Ocean Time Series just south of Australia (Yang et al., 2024) and is representative of open-ocean Southern Ocean waters. Over the continental shelf, moving from north to south, the amplitudes increase, with approximately an order of magnitude difference between the Drake Passage and the Southern Shelf subregion amplitudes (from 18–186 µatm to 0.5–4.5 mg m<sup>-3</sup>) (Figure 2, Table 1). While the Northern Shelf and Mid Shelf show strong relationships between Chl-a and the non-thermal component of  $pCO_{2sur}$ , the seasonal cycles of carbon uptake there are not as strongly biologically driven nor as strong a sink as in the Southern Shelf. During austral summer the Northern Shelf acts as a CO<sub>2</sub> sink, but at the annual timescale operates as a net source (Ito et al., 2018; Monteiro, Kerr, Orselli, & Lencina-Avila, 2020). The Southern Shelf amplitudes are considerably large, demonstrating highly productive waters with some of the highest seasonal biologically driven inorganic carbon drawdown anywhere in the global ocean (Legge et al., 2015; Roobaert et al., 2019). The biological contribution of phytoplankton production is more substantial near the seasonal sea ice zone, possibly due to sea ice's effects to enhance the stability of the upper water column, shielding from wind mixing, or ice edge upwelling, or gradients in nutrient availability (Smith, 1987), drivers which require future research.

#### 4.3. Ecosystem Implications

Changes in carbon uptake are intricately associated with shifts in the overall carbonate chemistry of the system, occurring simultaneously and influencing each other through various feedback mechanisms. There may be a shift in pH and saturation state of aragonite especially in summer and fall over the shelf. In the Australian sector of Southern Ocean, the seasonal cycle of  $pCO_2$  increased in amplitude from 2011 to 2020 with a corresponding decrease in pH (Shadwick et al., 2023). In the WAP, large biological carbon drawdown in summer generally increases aragonite saturation state in surface waters nearshore despite meltwater inputs, thus there are unlikely to be negative consequences in terms of acidified conditions for calcifying organisms in this system in the near term. In terms of phytoplankton community composition, diatoms are particularly important for the efficiency of the biological pump over the shelf in this region (Brown et al., 2019; Costa et al., 2020), and the seasonal succession of different groups and top-down grazing pressure likely moderate the biological contribution to  $\Delta pCO_2$  (Costa et al., 2023); these factors merit further study. Overall, our results highlight the need for a broader understanding of seasonal variability and long-term trends in the WAP marine ecosystem, including continued monitoring and research efforts to assess the impacts of climate change on coastal Antarctic environments.

### 5. Conclusions

In summary, the seasonality of surface ocean carbon uptake varies with latitude in the WAP region, and seasonal cycles show strong biologically driven uptake moving poleward. Seasonal cycles of both  $\Delta pCO_2$  and Chl-a increase in amplitude moving poleward, from small-amplitude seasonal cycles (18 µatm and 0.5 mg m<sup>-3</sup>) in

the Drake Passage to high-amplitude seasonal cycles (186  $\mu$ atm and 4.5 mg m<sup>-3</sup>) over the southern continental shelf region. Massive biological drawdown of CO<sub>2</sub> occurs over the continental shelf in summer, with biological utilization driving pCO<sub>2sur</sub> substantially lower, overshadowing the tendency for summer warming and low solubility to concurrently raise pCO<sub>2sur</sub>. These findings help answer whether the Southern Ocean acts as a net sink or source of CO<sub>2</sub> in the context of the global carbon budget.

## **Appendix A: Notation**

Abbreviation Definition Amplitude Maximum minus minimum monthly values over the seasonal cycle Chl-a Chlorophyll-a pigment concentration (mg  $m^{-3}$ ) OC-CCI Ocean color climate change initiative OC2 Ocean Color-2 Chl-a algorithm using two wavelengths (O'Reilly et al., 2000) OCI Ocean Color Index Chl-a algorithm (Hu et al., 2019) OCI2 Chl-a algorithm adjusted as a hybrid between one version of OCI and OCX (Hu et al., 2019). OCX Ocean Color-3 to -6 series of Chl-a algorithms using 3 to 6 wavelengths (O'Reilly & Werdell, 2019).  $\Delta pCO_2$ Air-sea CO<sub>2</sub> flux, the difference pCO<sub>2sur</sub>-pCO<sub>2atm</sub> (µatm) pCO<sub>2atm</sub> Partial pressure of carbon dioxide in the atmosphere (µatm) pCO<sub>2sur</sub> Partial pressure of carbon dioxide in the surface ocean (µatm) pCO<sub>2sur</sub>-T Thermal component of the pCO<sub>2sur</sub> seasonal cycle (µatm) pCO<sub>2sur</sub>-non-T Non-thermal component of the pCO<sub>2sur</sub> seasonal cycle (µatm) Pal-LTER Palmer Long-term ecological research program SOCAT Surface Ocean CO<sub>2</sub> Atlas SST Sea surface temperature WAP West Antarctic Peninsula

## **Data Availability Statement**

Ocean color data used in satellite-derived Chl-a analysis are available from the Ocean Colour Climate Change Initiative (OC-CCI) (Sathyendranath et al., 2019) and are accessible at https://www.oceancolour.org/thredds/catalog-cci.html?dataset=CCI\_ALL-v6.0-MONTHLY. Surface ocean  $pCO_2$  data are available from the Surface Ocean CO<sub>2</sub> Atlas (SOCAT) v2023 (Bakker et al., 2016) and are accessible at https://socat.info/index.php/data-access/. For this study SOCAT V2023 was used which is available at https://socat.info/index.php/previous-versions/.  $pCO_{2atm}$  is the dry air mixing ratio of atmospheric CO<sub>2</sub> (xCO<sub>2</sub>) from the Global Monitoring Laboratory surface marine boundary layer CO<sub>2</sub> product available at https://gml.noaa.gov/ccgg/mbl/mbl.html (last access: 19 November 2024) (Lan et al., 2023). Code to make figures in this paper is available at https://doi.org/10.5281/zenodo.14608738 (Turner, 2025).

19448007, 2025, 4, Downloaded from http:



#### Acknowledgments

This study was supported by an NSF Office of Polar Programs Postdoctoral Research Fellowship to J. Turner award number 2317774. Additional funding comes from Palmer Long Term Ecological Research Grant NSF LTER-2026045 and NASA Interdisciplinary Research in Earth Sciences (IDS) award number 80NSSC24K0481. The Surface Ocean CO2 Atlas (SOCAT) is an international effort, endorsed by the International Ocean Carbon Coordination Project (IOCCP), the Surface Ocean Lower Atmosphere Study (SOLAS) and the Integrated Marine Biosphere Research (IMBeR) program, to deliver a uniformly quality-controlled surface ocean CO2 database. The many researchers and funding agencies responsible for the collection of data and quality control are thanked for their contributions to SOCAT. This paper was improved by comments from two anonymous reviewers.

### References

- Arbilla, L. A., Ruiz-Etcheverry, L. A., López-Abbate, C., & Kahl, L. C. (2024). CO<sub>2</sub> sink and source zones delimited by marine fronts in the Drake Passage. Progress in Oceanography, 223, 103246. https://doi.org/10.1016/j.pocean.2024.103246
- Arrigo, K. R., van Dijken, G., & Long, M. (2008). Coastal Southern Ocean: A strong anthropogenic CO<sub>2</sub> sink. *Geophysical Research Letters*, 35(21), L21602. https://doi.org/10.1029/2008GL035624
- Arroyo, M. C., Shadwick, E. H., Tilbrook, B., Rintoul, S. R., & Kusahara, K. (2020). A continental shelf pump for CO<sub>2</sub> on the Adélie Land coast, East Antarctica. *Journal of Geophysical Research: Oceans*, 125(10). https://doi.org/10.1029/2020JC016302
- Bakker, D. C. E., Pfeil, B., Landa, C. S., Metzl, N., O'Brien, K. M., Olsen, A., et al. (2016). A multi-decade record of high-quality fCO<sub>2</sub> data in version 3 of the Surface Ocean CO<sub>2</sub> Atlas (SOCAT). *Earth System Science Data*, 8(2), 383–413. https://doi.org/10.5194/essd-8-383-2016
- Bernardello, R., Marinov, I., Palter, J. B., Galbraith, E. D., & Sarmiento, J. L. (2014). Impact of Weddell Sea deep convection on natural and anthropogenic carbon in a climate model. *Geophysical Research Letters*, 41(20), 7262–7269. https://doi.org/10.1002/2014GL061313
- Brown, M. S., Munro, D. R., Feehan, C. J., Sweeney, C., Ducklow, H. W., & Schofield, O. M. (2019). Enhanced oceanic CO<sub>2</sub> uptake along the rapidly changing West Antarctic Peninsula. *Nature Climate Change*, 9(9), 678–683. https://doi.org/10.1038/s41558-019-0552-3
- Bushinsky, S. M., Landschützer, P., Rödenbeck, C., Gray, A. R., Baker, D., Mazloff, M. R., et al. (2019). Reassessing Southern Ocean air-sea CO<sub>2</sub> flux estimates with the addition of biogeochemical float observations. *Global Biogeochemical Cycles*, 33(11), 1370–1388. https://doi.org/10. 1029/2019GB006176
- Caldeira, K., & Duffy, P. B. (2000). The role of the Southern Ocean in uptake and storage of anthropogenic carbon dioxide. *Science*, 287(5453), 620–622. https://doi.org/10.1126/science.287.5453.620
- Carrillo, C. J., Smith, R. C., & Karl, D. M. (2004). Processes regulating oxygen and carbon dioxide in surface waters West of the Antarctic Peninsula. *Marine Chemistry*, 84(3–4), 161–179. https://doi.org/10.1016/j.marchem.2003.07.004
- Cimino, M. A., Conroy, J. A., Connors, E., Bowman, J., Corso, A., Ducklow, H., et al. (2023). Long-term patterns in ecosystem phenology near Palmer Station, Antarctica, from the perspective of the Adélie Penguin. *Ecosphere*, *14*(2), e4417. https://doi.org/10.1002/ecs2.4417
- Costa, R. R., Ferreira, A., De Souza, M. S., Tavano, V. M., Kerr, R., Secchi, E. R., et al. (2023). Physical-biological drivers modulating phytoplankton seasonal succession along the Northern Antarctic Peninsula. *Environmental Research*, 231, 116273. https://doi.org/10.1016/j. envres.2023.116273
- Costa, R. R., Mendes, C. R. B., Tavano, V. M., Dotto, T. S., Kerr, R., Monteiro, T., et al. (2020). Dynamics of an intense diatom bloom in the Northern Antarctic Peninsula, February 2016. *Limnology & Oceanography*, 65(9), 2056–2075. https://doi.org/10.1002/lno.11437
- Dickson, A. G., Sabine, C. L., & Christian, J. R. (Eds.) (2007). IOCCP Report 8, *Guide to best practices for ocean CO*<sub>2</sub> measurements (Vol. 3). PICES Special Publication.
- Dierssen, H. (2000). Ocean color remote sensing of chlorophyll and primary production west of the Antarctic Peninsula Ph.D. Dissertation. University of California.
- Dierssen, H., & Smith, R. C. (2000). Bio-optical properties and remote sensing ocean color algorithms for Antarctic Peninsula waters. Journal of Geophysical Research, 105(C11), 26301–26312. https://doi.org/10.1029/1999JC000296
- Ducklow, H. W., Fraser, W. R., Meredith, M. P., Stammerjohn, S. E., Doney, S. C., Martinson, D. G., et al. (2013). West Antarctic peninsula: An ice-dependent coastal marine ecosystem in transition. *Oceanography*, 26(3), 190–203. https://doi.org/10.5670/oceanog.2013.62
- Eveleth, R., Cassar, N., Doney, S. C., Munro, D. R., & Sweeney, C. (2017). Biological and physical controls on O<sub>2</sub>/Ar, Ar and pCO<sub>2</sub> variability at the Western Antarctic Peninsula and in the Drake Passage. *Deep Sea Research Part II: Topical Studies in Oceanography*, 139, 77–88. https:// doi.org/10.1016/j.dsr2.2016.05.002
- Fay, A. R., Lovenduski, N. S., McKinley, G. A., Munro, D. R., Sweeney, C., Gray, A. R., et al. (2018). Utilizing the Drake passage time-series to understand variability and change in subpolar Southern Ocean. *Biogeosciences*, 15(12), 3841–3855. https://doi.org/10.5194/bg-15-3841-2018
- Fay, A. R., & McKinley, G. A. (2014). Global open-ocean biomes: Mean and temporal variability. Earth System Science Data, 6(2), 273–284. https://doi.org/10.5194/essd-6-273-2014
- Fay, A. R., & McKinley, G. A. (2017). Correlations of surface ocean pCO<sub>2</sub> to satellite chlorophyll on monthly to interannual timescales. *Global Biogeochemical Cycles*, 31(3), 436–455. https://doi.org/10.1002/2016GB005563
- Friedlingstein, P., O'Sullivan, M., Jones, M. W., Andrew, R. M., Bakker, D. C. E., Hauck, J., et al. (2023). Global carbon budget 2023. Earth System Science Data, 15(12), 5301–5369. https://doi.org/10.5194/essd-15-5301-2023
- Gohin, F., Druon, J. N., & Lampert, L. (2002). A five channel Chlorophyll concentration algorithm applied to SeaWiFS data processed by SeaDAS in coastal waters. *International Journal of Remote Sensing*, 23(8), 1639–1661. https://doi.org/10.1080/01431160110071879
- Gray, A. R., Johnson, K. S., Bushinsky, S. M., Riser, S. C., Russell, J. L., Talley, L. D., et al. (2018). Autonomous biogeochemical floats detect significant carbon dioxide outgassing in the high-latitude Southern Ocean. *Geophysical Research Letters*, 45(17), 9049–9057. https://doi.org/ 10.1029/2018GL078013
- Haëntjens, N., Boss, E., & Talley, L. D. (2017). Revisiting ocean color algorithms for chlorophyll a and particulate organic carbon in the Southern Ocean using biogeochemical floats. Journal of Geophysical Research: Oceans, 122(8), 6583–6593. https://doi.org/10.1002/2017JC012844
- Hauri, C., Doney, S. C., Takahashi, T., Erickson, M., Jiang, G., & Ducklow, H. W. (2015). Two decades of inorganic carbon dynamics along the West Antarctic Peninsula. *Biogeosciences*, 12(22), 6761–6779. https://doi.org/10.5194/bg-12-6761-2015
- Henley, S. F., Schofield, O. M., Hendry, K. R., Schloss, I. R., Steinberg, D. K., Moffat, C., et al. (2019). Variability and change in the west Antarctic Peninsula marine system: Research priorities and opportunities. *Progress in Oceanography*, 173, 208–237. https://doi.org/10.1016/j. pocean.2019.03.003
- Hu, C., Feng, L., Lee, Z., Franz, B. A., Bailey, S. W., Werdell, P. J., & Proctor, C. W. (2019). Improving satellite global chlorophyll a data products through algorithm refinement and data recovery. *Journal of Geophysical Research: Oceans*, 124, 1524–1543. https://doi.org/10.1029/ 2019JC014941
- Hu, C., Lee, Z., & Franz, B. (2012). Chlorophyll a algorithms for oligotrophic oceans: A novel approach based on three-band reflectance difference. *Journal of Geophysical Research*, 117(1), C01011. https://doi.org/10.1029/2011JC007395
- Ito, R. G., Tavano, V. M., Mendes, C. R. B., & Garcia, C. A. E. (2018). Sea-air CO<sub>2</sub> fluxes and pCO<sub>2</sub> variability in the Northern Antarctic Peninsula during three summer periods (2008–2010). Deep Sea Research Part II: Topical Studies in Oceanography, 149, 84–98. https://doi. org/10.1016/j.dsr2.2017.09.004
- Kahru, M., & Mitchell, B. G. (2010). Blending of ocean colour algorithms applied to the Southern Ocean. *Remote Sensing Letters*, 1(2), 119–124. https://doi.org/10.1080/01431160903547940
- Kalnay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L., et al. (1996). The NCEP/NCAR 40-Year reanalysis Project. Bulletin of the American Meteorological Society, 77(3), 437–471. https://doi.org/10.1175/1520-0477(1996)077<0437:TNYRP>2.0.CO;2

- Kim, H. H., Ducklow, H. W., Abele, D., Barlett, E. M. R., Buma, A. G. J., Meredith, M. P., et al. (2018). Inter-decadal variability of phytoplankton biomass along the coastal West Antarctic Peninsula. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences*, 376(2122), 20170174. https://doi.org/10.1098/rsta.2017.0174
- Körtzinger, A., Send, U., Lampitt, R. S., Hartman, S., Wallace, D. W. R., Karstensen, J., et al. (2008). The seasonal pCO<sub>2</sub> cycle at 49°N/16.5°W in the northeastern Atlantic Ocean and what it tells us about biological productivity. *Journal of Geophysical Research*, 113(C4), 2007JC004347. https://doi.org/10.1029/2007JC004347
- Lan, X., Tans, P., & Thoning, K., & NOAA Global Monitoring Laboratory. (2023). NOAA greenhouse gas marine boundary layer reference— CO, [Dataset]. NOAA GML. https://doi.org/10.15138/DVNP-F961
- Legge, O. J., Bakker, D. C. E., Johnson, M. T., Meredith, M. P., Venables, H. J., Brown, P. J., & Lee, G. A. (2015). The seasonal cycle of oceanatmosphere CO<sub>2</sub> flux in Ryder Bay, West Antarctic Peninsula. *Geophysical Research Letters*, 42(8), 2934–2942. https://doi.org/10.1002/ 2015GL063796
- Legge, O. J., Bakker, D. C. E., Meredith, M. P., Venables, H. J., Brown, P. J., Jones, E. M., & Johnson, M. T. (2017). The seasonal cycle of carbonate system processes in Ryder Bay, West Antarctic Peninsula. *Deep Sea Research Part II: Topical Studies in Oceanography*, 139, 167– 180. https://doi.org/10.1016/j.dsr2.2016.11.006
- Long, M. C., Stephens, B. B., McKain, K., Sweeney, C., Keeling, R. F., Kort, E. A., et al. (2021). Strong Southern Ocean carbon uptake evident in airborne observations. *Science*, 374(6572), 1275–1280. https://doi.org/10.1126/science.abi4355
- Lüger, H., Wallace, D. W. R., Körtzinger, A., & Nojiri, Y. (2004). The pCO<sub>2</sub> variability in the midlatitude North Atlantic Ocean during a full annual cycle. *Global Biogeochemical Cycles*, 18(3), 2003GB002200. https://doi.org/10.1029/2003GB002200
- Marinov, I., Gnanadesikan, A., Toggweiler, J. R., & Sarmiento, J. L. (2006). The Southern Ocean biogeochemical divide. Nature, 441(7096), 964– 967. https://doi.org/10.1038/nature04883
- Mitchell, B. G. (1992). Predictive bio-optical relationships for polar oceans and marginal ice zones. Journal of Marine Systems, 3(1–2), 91–105. https://doi.org/10.1016/0924-7963(92)90032-4
- Mitchell, B. G., & Holm-Hansen, O. (1991). Bio-optical properties of Antarctic Peninsula waters: Differentiation from temperate ocean models. *Deep-Sea Research, Part A: Oceanographic Research Papers*, 38(8–9), 1009–1028. https://doi.org/10.1016/0198-0149(91)90094-V
- Mo, A., Park, K., Park, J., Hahm, D., Kim, K., Ko, Y. H., et al. (2023). Assessment of austral autumn air–sea CO<sub>2</sub> exchange in the Pacific sector of the Southern Ocean and dominant controlling factors. Frontiers in Marine Science, 10, 1192959. https://doi.org/10.3389/fmars.2023.1192959
- Monteiro, T., Kerr, R., & Machado, E. D. C. (2020). Seasonal variability of net sea-air CO<sub>2</sub> fluxes in a coastal region of the northern Antarctic Peninsula. Scientific Reports, 10(1), 14875. https://doi.org/10.1038/s41598-020-71814-0
- Monteiro, T., Kerr, R., Orselli, I. B. M., & Lencina-Avila, J. M. (2020). Towards an intensified summer CO<sub>2</sub> sink behaviour in the Southern Ocean coastal regions. *Progress in Oceanography*, 183, 102267. https://doi.org/10.1016/j.pocean.2020.102267
- Munro, D. R., Lovenduski, N. S., Stephens, B. B., Newberger, T., Arrigo, K. R., Takahashi, T., et al. (2015). Estimates of net community production in the Southern Ocean determined from time series observations (2002–2011) of nutrients, dissolved inorganic carbon, and surface ocean pCO<sub>2</sub> in Drake Passage. *Deep Sea Research Part II: Topical Studies in Oceanography*, 114, 49–63. https://doi.org/10.1016/j.dsr2.2014. 12.014
- Munro, D. R., Lovenduski, N. S., Takahashi, T., Stephens, B. B., Newberger, T., & Sweeney, C. (2015b). Recent evidence for a strengthening CO<sub>2</sub> sink in the Southern Ocean from carbonate system measurements in the Drake Passage (2002–2015). *Geophysical Research Letters*, 42(18), 7623–7630. https://doi.org/10.1002/2015GL065194
- O'Reilly, J. E., Maritorena, S., Siegel, D. A., O'Brien, M. C., Toole, D., Mitchell, B. G., et al. (2000). Ocean color chlorophyll a algorithms for SeaWiFS, OC2, and OC4: Version 4. SeaWiFS postlaunch calibration and validation analyses, Part 3 (pp. 9–23).
- O'Reilly, J. E., & Werdell, P. J. (2019). Chlorophyll algorithms for ocean color sensors—OC4, OC5 & OC6. Remote Sensing of Environment, 229(April), 32–47. https://doi.org/10.1016/j.rse.2019.04.021
- Reid, P., Stammerjohn, S., Massom, R. A., Barreira, S., Scambos, T. A., & Lieser, J. L. (2024). Sea-ice extent, concentration, and seasonality [in "State of the Climate in 2023"]. Bulletin of the American Meteorological Society, 105(8), S21–S24. https://doi.org/10.1175/BAMS-D-24-0099.1
- Roobaert, A., Laruelle, G. G., Landschützer, P., Gruber, N., Chou, L., & Regnier, P. (2019). The spatiotemporal dynamics of the sources and sinks of CO<sub>2</sub> in the global coastal ocean. *Global Biogeochemical Cycles*, 33(12), 1693–1714. https://doi.org/10.1029/2019GB006239
- Sarmiento, J. L., & Orr, J. C. (1991). Three-dimensional simulations of the impact of Southern Ocean nutrient depletion on atmospheric CO<sub>2</sub> and ocean chemistry. *Limnology & Oceanography*, 36(8), 1928–1950. https://doi.org/10.4319/10.1991.36.8.1928
- Sathyendranath, S., Brewin, R., Brockmann, C., Brotas, V., Calton, B., Chuprin, A., et al. (2019). An ocean-colour time series for use in climate studies: The experience of the Ocean-Colour Climate Change Initiative (OC-CCI). *Sensors*, *19*(19), 4285. https://doi.org/10.3390/s19194285
- Schofield, O., Brown, M., Kohut, J., Nardelli, S., Saba, G., Waite, N., & Ducklow, H. (2018). Changes in the upper ocean mixed layer and phytoplankton productivity along the West Antarctic Peninsula. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences*, 376(2122), 20170173. https://doi.org/10.1098/rsta.2017.0173
- Schofield, O., Saba, G., Coleman, K., Carvalho, F., Couto, N., Ducklow, H., et al. (2017). Decadal variability in coastal phytoplankton community composition in a changing West Antarctic Peninsula. Deep-Sea Research Part I Oceanographic Research Papers, 124(April), 42–54. https:// doi.org/10.1016/j.dsr.2017.04.014
- Schultz, C., Doney, S. C., Hauck, J., Kavanaugh, M. T., & Schofield, O. (2021). Modeling phytoplankton blooms and inorganic carbon responses to sea-ice variability in the West Antarctic Peninsula. *Journal of Geophysical Research: Biogeosciences*, 126(4), 1–21. https://doi.org/10.1029/ 2020JG006227
- Shadwick, E. H., De Meo, O. A., Schroeter, S., Arroyo, M. C., Martinson, D. G., & Ducklow, H. (2021). Sea ice suppression of CO<sub>2</sub> outgassing in the West Antarctic peninsula: Implications for the evolving Southern Ocean carbon sink. *Geophysical Research Letters*, 48(11), e2020GL091835. https://doi.org/10.1029/2020GL091835
- Shadwick, E. H., Wynn-Edwards, C. A., Matear, R. J., Jansen, P., Schulz, E., & Sutton, A. J. (2023). Observed amplification of the seasonal CO<sub>2</sub> cycle at the Southern Ocean Time series. *Frontiers in Marine Science*, 10, 1281854. https://doi.org/10.3389/fmars.2023.1281854
- Smith, W. O. (1987). Phytoplankton dynamics in marginal ice zones. In H. Barnes & M. Barnes (Eds.), Oceanography and marine biology. An annual review (Vol. 25, pp. 1–25). Aberdeen University Press.
  Stammarian S. E., & Sambas, T. A. (2020). Warming resolves the south Pala. Nature Climate Change, 10(8), 710, 711. https://doi.org/10.1038/
- Stammerjohn, S. E., & Scambos, T. A. (2020). Warming reaches the south Pole. Nature Climate Change, 10(8), 710–711. https://doi.org/10.1038/ s41558-020-0827-8
- St-Laurent, P., Yager, P. L., Sherrell, R. M., Oliver, H., Dinniman, M. S., & Stammerjohn, S. E. (2019). Modeling the seasonal cycle of iron and carbon fluxes in the Amundsen Sea Polynya, Antarctica. *Journal of Geophysical Research: Oceans*, 124(3), 1544–1565. https://doi.org/10. 1029/2018JC014773

- Takahashi, T., Olafsson, J., Goddard, J. G., Chipman, D. W., & Sutherland, S. C. (1993). Seasonal variation of CO<sub>2</sub> and nutrients in the highlatitude surface oceans: A comparative study. *Global Biogeochemical Cycles*, 7(4), 843–878. https://doi.org/10.1029/93GB02263
- Takahashi, T., Sutherland, S. C., Sweeney, C., Poisson, A., Metzl, N., Tilbrook, B., et al. (2002). Global sea–air CO<sub>2</sub> flux based on climatological surface ocean pCO<sub>2</sub>, and seasonal biological and temperature effects. *Deep Sea Research Part II: Topical Studies in Oceanography*, 49(9–10), 1601–1622. https://doi.org/10.1016/S0967-0645(02)00003-6
- Testa, G., Piñones, A., & Castro, L. R. (2021). Physical and biogeochemical regionalization of the Southern Ocean and the CCAMLR zone 48.1. *Frontiers in Marine Science*, 8, 592378. https://doi.org/10.3389/fmars.2021.592378
- Thomalla, S. J., Fauchereau, N., Swart, S., & Monteiro, P. M. S. (2011). Regional scale characteristics of the seasonal cycle of Chlorophyll in the Southern Ocean. *Biogeosciences*, 8(10), 2849–2866. https://doi.org/10.5194/bg-8-2849-2011
- Thomalla, S. J., Nicholson, S.-A., Ryan-Keogh, T. J., & Smith, M. E. (2023). Widespread changes in Southern Ocean phytoplankton blooms linked to climate drivers. *Nature Climate Change*, 13(9), 975–984. https://doi.org/10.1038/s41558-023-01768-4
- Tortell, P. D., Bittig, H. C., Körtzinger, A., Jones, E. M., & Hoppema, M. (2015). Biological and physical controls on N<sub>2</sub>, O<sub>2</sub>, and CO<sub>2</sub> distributions in contrasting Southern Ocean surface waters. *Global Biogeochemical Cycles*, 29(7), 994–1013. https://doi.org/10.1002/2014GB004975
- Turner, J. S. (2025). A data repository for: Seasonal variability of surface ocean carbon uptake and chlorophyll-a concentration in the West Antarctic Peninsula over two decades [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.14608738
- Turner, J. S., Dierssen, H., Schofield, O., Kim, H. H., Stammerjohn, S., Munro, D. R., & Kavanaugh, M. (2024). Changing phytoplankton phenology in the marginal ice zone west of the Antarctic Peninsula. *Marine Ecology Progress Series*, 734, 1–21. https://doi.org/10.3354/ meps14567
- Wanninkhof, R., Pierrot, D., Sullivan, K., Mears, P., & Barbero, L. (2022). Comparison of discrete and underway CO<sub>2</sub> measurements: Inferences on the temperature dependence of the fugacity of CO<sub>2</sub> in seawater. *Marine Chemistry*, 247, 104178. https://doi.org/10.1016/j.marchem.2022. 104178
- Yang, B., Shadwick, E. H., Schultz, C., & Doney, S. C. (2021). Annual mixed layer carbon budget for the West Antarctic Peninsula continental shelf: Insights from year-round mooring measurements. *Journal of Geophysical Research: Oceans*, 126(4), e2020JC016920. https://doi.org/10. 1029/2020JC016920
- Yang, X., Wynn-Edwards, C. A., Strutton, P. G., & Shadwick, E. H. (2024). Drivers of air-sea CO<sub>2</sub> flux in the Subantarctic Zone revealed by time series observations. *Global Biogeochemical Cycles*, 38(1), e2023GB007766. https://doi.org/10.1029/2023GB007766